Author’s response to reviews

Title: Post Traumatic Stress Disorder and Coping Strategies among Adult Survivors of Earthquake, Nepal

Authors:

Ishwari Adhikari Baral (adhikari.ishwari@gmail.com)

Version: 2 Date: 25 Jan 2019

Author’s response to reviews:

Hissei Imai

BMC Psychiatry

Dear Hissei Imai, regarding the english language improvements, I have done best possible revision with the help from two journal editors here in Nepal, as well as I have made major correction with the help of BMC writing resources. I hope this has made the manuscript better to be published in your Journal. I am hoping for the positive response.